Author's response to reviews

Title: Severe refractory autoimmune hemolytic anemia with both warm and cold autoantibodies: complete response to Rituximab

Authors:

Shilpi Gupta (drshilpigupta@gmail.com)
Anita Szerszen (aszerszen@aol.com)
Fadi Nakhl (fnakhl@hotmail.com)
Seema Varma (svarma@siuh.edu)
Aaron Gottesman (agottesman@siuh.edu)
Frank Forte (fforte@siuh.edu)
Meekoo Dhar (mdhar@siuh.edu)

Version: 2 Date: 18 April 2010

Author's response to reviews: see over
Dear Editor

Journal of Medical Case Reports

London

Subject: Revised manuscript

Thank you for reviewing our paper and we apologize for the errors in the manuscript. All the recommended changes have been made and incorporated in our manuscript. We have added the abstract section with the suggested headings. Ethnicity of the patient has been mentioned and conclusion section added at the end of the paper. We have also added an informed consent, competing interests, authors’ contribution and legend section.

We look forward to the final decision regarding our paper.

Thank you

Shilpi Gupta